Author's response to reviews

Title: Systemic Tuberculosis Presenting with Acute Transient Myopia: a case report

Authors:

Sher A Aslam (Sher.Aslam@moorfields.nhs.uk)
Shahram Kashani (shahdoc@hotmail.com)
Roland K Morley (rkmorley@gmail.com)

Version: 2 Date: 30 March 2008

Author's response to reviews:

Thank you very much for allowing the revisions which are as follows:
1. Restructured title "Systemic Tuberculosis Presenting with Acute Transient Myopia: a case".
2. Patient's ethnic origin in case presentation section of abstract included.
3. No abbreviations used in the abstract.
4. List of abbreviations used in the manuscript and their meanings included.
5. 'Competing interests' heading added.